Author's response to reviews

Title: Implementation and evaluation of a nurse-centered computerized potassium regulation protocol in the intensive care unit: a before and after analysis.

Authors:

Miriam Hoekstra (m.hoekstra@thorax.umcg.nl)
Mathijs Vogelzang (m.vogelzang@thorax.umcg.nl)
Jose T Drost (j.t.drost@student.rug.nl)
Marcel Janse (m.janse@student.rug.nl)
Bert G Loef (b.g.loef@thorax.umcg.nl)
Iwan CC van der Horst (i.van.der.horst@thorax.umcg.nl)
Felix Zijlstra (f.zijlstra@thorax.umcg.nl)
Maarten WN Nijsten (m.w.n.nijsten@chir.umcg.nl)

Version: 4  Date: 18 January 2010

Author's response to reviews:

Melissa Norton, MD
Editor-in-Chief
BMC Medical Informatics and Decision Making
Groningen, January 18th 2010

Dear Dr. Melissa Norton,

Thank you for the opportunity to publish our work in "BMC Medical Informatics and Decision Making".

We hereby submit a revised version of our manuscript entitled “Implementation and evaluation of a nurse-centered computerized potassium regulation protocol in the intensive care unit: a before and after analysis”.

We added a statement concerning the ethical approval of our local review board, with a reference number.

If there are any questions left, please do not hesitate to contact us.

Yours sincerely, on behalf of all authors,

Miriam Hoekstra
Mathijs Vogelzang
Maarten Nijsten
Please address correspondence to:

Miriam Hoekstra, MD
University Medical Center Groningen, Department of Cardiology
PO 30.001, Hanzeplein 1, 9700 RB Groningen, The Netherlands
Email: m.hoekstra@thorax.umcg.nl
Tel: 0031-50-3611413
Fax: 0031-50-3611347